Author's response to reviews

Title: Perinatal HIV transmission and the cost-effectiveness of screening at 14 weeks gestation, at the onset of labour and the rapid testing of infants.

Authors:

Belinda Udeh (belinda-udeh@uiowa.edu)
Chiedozie Udeh (n.graves@qut.edu.au)
Nicholas Graves (n.graves@qut.edu.au)

Version: 3 Date: 28 November 2008

Author's response to reviews:

November 26th, 2008
Melissa Norton MD, Editor (Medicine)
BMC Infectious Diseases
Biomed Central Ltd.
Science Navigation Group
Middlesex House
34-42 Cleveland St.
London W1T 4LD, UK

Dear Dr. Norton,

Thank you for acceptance of the publication of our article, ‘Perinatal HIV transmission and the cost-effectiveness of screening at 14 weeks gestation, at the onset of labour and the rapid testing of infants’ within your journal.

We appreciate the efforts made by your reviewers and editing staff. We have attached the edited manuscript. As recommended, we have addressed and improved the manuscript based on the additional suggestions from the 2nd round of comments from your reviewers and editing staff.

Thank you again for the time and effort by yourself, editorial board and reviewers and thank you for the acceptance of our submission for publication.

On behalf of the authors,

Nicholas Graves, PhD
IHBI
Queensland University of Technology
Brisbane, Australia